Author's response to reviews

Title: Current status of collaborative relationships between dialysis facilities and dental facilities in Japan: results of a nationwide survey

Authors:

Masami Yoshioka (yoshioka.masami@tokushima-u.ac.jp)
Yasuhiko Shirayama (shirayama@tokushima-u.ac.jp)
Issei Imoto (issehgen@tokushima-u.ac.jp)
Daisuke Hinode (hinode@tokushima-u.ac.jp)
Shizuko Yanagisawa (yanagisawa@tokushima-u.ac.jp)
Yuko Takeuchi (yukotakeuchi@tokushima-u.ac.jp)

Version: 2
Date: 21 August 2014

Author's response to reviews:

Dear Editor,

Thank you for your cooperation.
I have revised our manuscript as follows:

1. Consent
I added the sentence as described below.
In this study, written informed consent for participation was not obtained from participants, because we considered the reply as the agreement with participation, as explained in the document which we sent with questionnaire to each facility.

2. Line numbering
According to your advice, I deleted insert page breaks in our manuscript.

3. Tables as additional files
We included tables in the manuscript since we want to the tables to be visible within the final published manuscript.
I would appreciate your cooperation.

With best regards,
Masami Yoshioka